Author’s response to reviews

Title: Antimicrobial characteristics of Berberine against Prosthetic Joint Infection-related Staphylococcus aureus of Different Multi-locus Sequence Types.

Authors:

Jiaqi Tan (tan984572523@163.com)
Jin Wang (wangjin6hao@gmail.com)
Chuang Yang (arthroyang@163.com)
Chongzun Zhu (zhuchongzun998@126.com)
Geyong Guo (1013368129@qq.com)
Jin Tang (tangjin6ph@163.com)
Hao Shen (shenhao7212@sina.com)

Version: 1 Date: 07 Dec 2018

Author’s response to reviews:

Thank you for reviewing and kind suggestion, which we have followed in our revision.